Author's response to reviews

Title: Molecular Epidemiology of Hepatitis B Virus Genotypes in Pakistan

Authors:

Muhammad M Alam (urmasroor@yahoo.com)
Sohail Z Zaidi (zaidiss@whopak.org)
Salman A Malik (salmanakber@yahoo.com)
Shahzad Shaukat (vibgyorss@yahoo.com)
Asif Naeem (drasifnaeemm@yahoo.com)
Salmaan Sharif (Salmaanshariff@yahoo.com)
Mehar Angez (meharangezz@yahoo.com)
Javed A Butt (javedaslam@yahoo.com)
Anis Khan (aniskhan@yahoo.com)

Version: 2 Date: 6 July 2007

Author's response to reviews: see over
The following revisions have been made in the revised submission according to the recommendations by reviewers.

1- Background of genotypes epidemiology has been revised.
2- Tables have been added to describe the demographics (geographical distribution, age, gender)
3- The figures have been omitted on reviewer’s recommendation.